Author’s response to reviews

Title: Liver fibrosis and accelerated immune dysfunction (immunosenescence) among HIV-infected Russians with heavy alcohol consumption - An observational cross-sectional study

Authors:
Kaku So-Arman (kaku@bu.edu)
Matthew Freiberg (matthew.s.freiberg@vanderbilt.edu)
Debbie Cheng (dmcheng@bu.edu)
Joseph Lim (joseph.lim@yale.edu)
Natalia Gnatienko (Natalia.Gnatienko@bmc.org)
Gregory Patts (gpatts@bu.edu)
Margaret Doyle (Margaret.Doyle@uvm.edu)
Daniel Fuster (daniel.fuster@bmc.org)
Dmitry Lioznov (dlioznov@ya.ru)
Evgeny Krupitsky (kruenator@gmail.com)
Jeffrey Samet (Jeff.Samet@bmc.edu)

Version: 3 Date: 02 Dec 2019

Author’s response to reviews:

Please find below, responses to reviewer and editor comments:

1. Permissions

Please clarify in the Methods section how access was granted to the ARCH/ZINC HIV data, or if access was not needed - for instance, if the data was publicly available as national archived data.

The datasets used and/or analyzed during the current study are available from the corresponding author on reasonable request.

2. Funding

--Please move the Funding subsection into the Declarations section
Completed as requested

--Please note the role(s) of the funding body (bodies) in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript, which should be declared in the Funding section.

Completed as requested

3. Author initials

In the “Authors’ contributions” section of the Declarations, please ensure initials match their provided names in the author list for this manuscript, in the correct order. For instance, an author named John Steven Doe, their initials should be represented as JSD in the “Authors’ contributions”; an author named Jane Ann Doe should be referred to as JAD.

Completed as requested

4. Authors’ contributions - approval

Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript in its current state.

Completed as requested

5. Formatting

In order to be in line with the submission guidelines for BMC Gastroenterology research articles (found in detail here: https://bmcgastroenterol.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article), please make the following formatting changes:

-Please rename the section titled Introduction to Background

Completed as requested

6. Remove supplementary file

Please remove the supplementary files "BMCGE response to reviewers v2.0.docx” from the manuscript, as these files are not required to be attached to the manuscript at this stage of the editorial process.

Completed as requested

7. Clean manuscript
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Completed as requested